Author’s response to reviews

Title: Indocyanine green angiography findings in patients with long-standing Vogt-Koyanagi-Harada disease: a cross-sectional study

Authors:

Felipe T da Silva MD (felipeg2000@yahoo.com)
Carlos E Hirata MD (carloshirata@uol.com.br)
Viviane M Sakata MD (ymsakata@yahoo.com.br)
Edilberto Olivalves MD (eolivalves@gmail.com)
Rony C Preti MD (rypreti@hotmail.com)
Sergio LG Pimentel MD (sergiopimentel9@gmail.com)
Andre V Gomes MD (andremv@uol.com.br)
Walter Y Takahashi MD (wytakahashi@uol.com.br)
Rogerio A Costa MD (rog.retina@uol.com.br)
Joyce H Yamamoto MD (joycehy@uol.com.br)

Version: 2 Date: 23 March 2012

Author’s response to reviews: see over
São Paulo, January 22th 2012.

Jigisha Patel, M.D.
Chief Editor
BMC Ophthalmology

Dear Dr. PATEL,

My colleagues and I are submitting our current research entitled “Indocyanine green angiography findings in patients with long-standing Vogt-Koyanagi-Harada disease: a cross-sectional study” for publication as full-length article in the BMC Ophthalmology. This research is an important part of our current research on Vogt-Koyanagi-Harada disease since, at the time this study was conducted, disease activity and therapy were solely monitored based on clinical and fluorescein angiographic features. This paper reinforces that monitorization of disease activity by clinical examination and fluorescein angiography are not adequate and reinforces the importance of indocyanine green angiography in monitoring disease activity and in better conducting treatment.

We thank you very much in advance.

Sincerely yours,

Joyce H Yamamoto et al